Author's response to reviews

Title: Non-infections osteomyelitis of the mandible of a young female: a case report

Authors:

Anne Q Rasmussen (anne.qvist.01@regionh.dk)
Ulrik B Andersen (ulrik.b.andersen@regionh.dk)
Niklas R Jørgensen (niklas@dadbnet.dk)
Peter Schwarz (peter.schwarz@regionh.dk)

Version: 5 Date: 5 December 2013

Author's response to reviews: see over
Dear Ms Vidal,

Thank you for the recent mail.

We have addressed the following:

1. Please include all the co-authors’ email addresses in the Title page.

   Done

2. Please reformat the Consent statement. We recommend the following wording is used for the consent section:

   "Written informed consent was obtained from the patient’s legal guardian(s) for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

   Done

Best regards

Peter Schwarz
Professor, MD, DMSci.
Director of Research Center of Ageing and Osteoporosis,
Department of Medicine M,
Glostrup Hospital, University of Copenhagen
2600 Glostrup
Phone +45 3863 4595 (direct) and phone +45 3863 3090 (secretary)
e-mail: peter.schwarz@regionh.dk